

Format for ANSWERING REVIEWERS

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 002429-review.doc).

Title: Indoor smoke and prenatal and childhood growth: The role of (gestational) age

Author: Rakesh Ghosh

Name of Journal: *World Journal of Clinical Pediatrics*

ESPS Manuscript NO: 2377

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Clinical Pediatrics*.